Author's response to reviews

Title: Clinical Features and Phylogenetic Tree of Coxsackievirus A9 in Northern Taiwan in 2011

Authors:

Yi-Chuan Huang (bb0479yc@gmail.com)
Ying-Hsia Chu (b95401099@ntu.edu.tw)
Ting-Yun Yen (ytingyu@gmail.com)
Wen-Chan Huang (summerdebbie@gmail.com)
Li-Min Huang (lmhuang@ntu.edu.tw)
Chang Chang (lychang@ntu.edu.tw)

Version: 2 Date: 3 July 2012

Author's response to reviews: see over
Dear Editor

MS: 9376959037492479
Research article
Clinical Features and Phylogenetic Tree of Coxsackievirus A9 in Northern Taiwan in 2011
Yi-Chuan Huang, Ying-Hsia Chu, Ting-Yun Yen, Wen-Chan Huang, Li-Min Huang and Chang Chang
BMC Infectious Diseases (Section: Viral diseases)

As you request to revise our manuscript, we make some changes to <the formatting of> our manuscript. We added in the method (Paragraph 1, Page 8):

“This study was approved by the Institutional Review Board of National Taiwan University Hospital and the approval number was 201007053R. Informed consent for participation in the study and the publication of the images was obtained from the parent of the enrolled children”.

Thank you very much for your consideration.

Sincerely yours,

Luan-Yin Chang, Department of Pediatrics, National Taiwan University Hospital,
College of Medicine, National Taiwan University, 8, Chung-Shan South Road, Taipei,
Taiwan 100.

Tel: 886-2-23123456 ext 71528
Fax: 886-2-23147450
E-mail: lychang@ntu.edu.tw or ly7077@tpts6.seed.net.tw